Author's response to reviews

Title: Obese adults attitudes towards individual and population based interventions for obesity: A qualitative community study

Authors:

Samantha L Thomas (samantha.thomas@med.monash.edu.au)
Sophie Lewis (sophie.lewis@med.monash.edu.au)
Jim Hyde (jim.hyde@dhs.vic.gov.au)
David Castle (David.CASTLE@svhm.org.au)
Paul Komesaroff (paul.komesaroff@med.monash.edu.au)

Version: 4 Date: 17 May 2010

Author's response to reviews:

Dear Editors,

Thank you for the further comments on this manuscript.

We have addressed the comments as follows and have also highlighted the changes in the manuscript itself:

1) Remove "different socio-economic backgrounds": We have removed this and have replaced with "...individuals with different experiences with their obesity"
2) Remove "and across" from the Data Analysis section of the paper: We have removed this.
3) Typographical error "necessary" should read "necessarily": This has been changed.
4) Grammatical error "in comprehensive approach": this has been changed to "as part of a comprehensive approach" (pg 19)

We have also endeavoured to fix the minor spacing error within the text.

Finally we have added a statement to the paper to specifically thank the reviewers Lily O'Hara and Dr Helen Mavoa, who provided such excellent reviews of the paper. I hope both reviewers are happy to be named in the acknowledgement section. If not, their names could be removed, and the thanks to the reviewers could remain.

Thank you again!

Best wishes,

Samantha Thomas